Author's response to reviews

Title: Cardiac safety of indacaterol in healthy subjects: a randomized, multidose, placebo- and positive-controlled, parallel-group thorough QT study

Authors:

Sanjeev Khindri (sanjeev.khindri@novartis.com)
Ronald Sabo (ron.sabo@novartis.com)
Stuart Harris (SHarris@SeaViewResearch.Net)
Ralph Woessner (ralph.woessner@novartis.com)
Simon Jennings (simon.jennings@novartis.com)
Anton F Drollmann (anton.drollmann@novartis.com)

Version: 3 Date: 16 May 2011

Author's response to reviews:

Dear Editorial Production Team,

As suggested for the final proofing stage of our manuscript, we have included ‘conclusion’ section after the discussion. Line numbers and track changes have been removed as well.